Author's response to reviews

Title: Internet-based medical education: A realist review of what works, for whom and in what circumstances

Authors:

- Geoff Wong (g.wong@ucl.ac.uk)
- Trisha Greenhalgh (p.greenhalgh@ucl.ac.uk)
- Ray Pawson (r.d.pawson@leeds.ac.uk)

Version: 6 Date: 26 January 2010

Author's response to reviews:

Dear Dr Norton,

Thank you for your email of 26th January 2010 asking us to make some formatting changes.

I am pleased to inform you that all your requests for Major and Minor revisions have been made and I hope what we have done meets with your approval.

Yours Sincerely

Dr Geoff Wong
For and behalf of all authors